Author's response to reviews

Title: The Lactobacillus flora in vagina and rectum of fertile and postmenopausal healthy Swedish women

Authors:

Rita J Gustafsson (rita.gustafsson@med.lu.se)
Siv Ahrne (siv.ahrne@appliednutrition.lth.se)
Bengt Jeppson (bengt.jeppson@med.lu.se)
Cecilia Benoni (cecilia.benoni@med.lu.se)
Crister Olsson (Crister.Olsson@med.lu.se)
Martin Stjernquist (martin.stjernquist@med.lu.se)
Bodil Ohlsson (bodil.ohlsson@med.lu.se)

Version: 5 Date: 16 May 2011

Author’s response to reviews:

Dear Editor, I have made the major revision by removing the tracking.

I have removed the vertical lines in the tables.

I have excluded the figure number in the figure files.

However, I have not been able to crop the figures. We are grateful to You if You can do this.

Sincerely Rita Gustafsson